Author’s response to reviews

Title: Characteristics and outcomes of patients with eclampsia and severe pre-eclampsia in a rural hospital in Western Tanzania: a retrospective medical record study

Authors:

Rob Mooij (r.mooij1983@gmail.com)
Joseph Lugumila (lugumila@gmail.com)
Masumbuko Y Mwashambwa (masuyt@yahoo.co.uk)
Ipyana H Mwampagatwa (imwampagatwa@yahoo.com)
Jeroen van Dillen (jeroen.vandillen1@radboudumc.nl)
Jelle Stekelenburg (jelle.stekelenburg@online.nl)

Version: 3 Date: 16 August 2014

Author’s response to reviews:

Dear Sir,

Hereby we would like to submit the 2nd revision of the above mentioned paper. We thank you for allowing us to give more clarification on the ethical consideration. We have sent the appropriate documents through e-mail.

We have changed some small details in the manuscript (typing errors, statistical data) and as requested we have amended the part about ethical consideration.

We are happy to address any other issues.

Yours sincerely,